Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
May 12, 2010
Injector    Energy 56.62 MeV
· (5/11) Wien angle was flipped back for Hall A during yesterday’s beam studies. Beam was not stable after the beam studies. Injector experts started investigation of the injector at 0830 today.

· (5/10) Injector was adjusted to make three beams coincide.

· (5/7)  Wien angle was flipped for Hall A.

· (5/7) The laser spot was moved to boost QE.
· (5/4)Some work on the Poekel cell was done to reduce the “ringing”.

· An injector study showed that the Wien filers may be the cause of the 5-second beam motion. The investigation is not complete yet.
Accelerator   Linac Energy 503.25 MeV
· Overall the accelerator was running well with low trip rates. Main machine optics has been stable.
· (5/11) In the current machine configuration, if BSY dipole is off ,  Hall C dumplette is out and Hall C beam is on, where does the beam go? 

· (5/10) Box Arc8 tripped off and Ops was able to reset it.

· (5/7) VBV1C19 would not open due to a disconnected air hose. When trying sending beam to Hall A the vacuum FSD node was masked. After changing the FSD destination if you have to mask something ask the question why do you need to do so. After CW beam is interrupted (by beam line work, magnet problems and etc.,or extended down) one should always send tune beam first to make sure beam is where it should be.
· (5/6) CHL tripped. The cause of the trip is still under investigation.

· (5/3) The accelerator optics was optimized, and Hall B and BSY were changed to pass 5. 

· (5/2) The CHL tripped possibly due to a power glitch, which caused more than five hours of downtime. 

· (5/1) After the open house the accelerator was recovered nicely.

· (4/30) Beam stopped for the open house.

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· Hall A still has issues with its target chamber vacuum.

· (5/11) Hall A started to take beam with beam going through the Compton chicane.

· (5/9) Hall A current lock was changed to slit mode. We do not know why yet.

· (5/6) Hall A target chamber vacuum went bad again.
· Hall A line optics was changed to enlarge the beam size (MQA1H01, and the Moller quads were set to zero field).

· (5/2) After the open house Hall A started to take straight-ahead mode beam due to the work on the Compton laser cavity mirrors.  

· (4/29) Hall A stopped taking beam to fix a vacuum leak downstream of the target chamber. 

Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· (5/11) Hall B beam was not stable after yesterday’s beam studies.

· (5/6) Hall B current and positions were not stable over the weekend. Ops cleaned up the injector (three beams did not coincide), orbits in the accelerator and Hall B line. Hall B beam was stabilized after the clean-up.
· (5/3) Hall B was changed to pass 5.

· No problems with Hall B beam

Hall C---QWeak from June 8 to August 15, 2010
· QWeak installation
